Author's response to reviews

Title: Core competencies in the science and practice of knowledge translation: description of a Canadian strategic training initiative

Authors:

Sharon E Straus (sharon.straus@utoronto.ca)
Melissa Brouwers (mbrouwer@mcmaster.ca)
David Johnson (david.johnson@albertahealthservices.ca)
John N Lavis (lavisj@mcmaster.ca)
France Légaré (france.legare@fmed.ulaval.ca)
Sumit R Majumdar (me2.majumdar@ualberta.ca)
K Ann McKibbon (mckib@mcmaster.ca)
Anne E Sales (anne.sales1@gmail.com)
Dawn Stacey (dawn.stacey@uottawa.ca)
Gail Klein (kleing@smh.ca)
Jeremy Grimshaw (jgrimshaw@ohri.ca)
KT Canada Strategic Training Initiative in Health Research STIHR (kleing@smh.ca)

Version: 6 Date: 2 December 2011

Author's response to reviews:

Revisions made as per your email dated November 28, 2011